Author’s response to reviews

Title: Heterogeneity of PD-L1 Expression in Primary Tumors and Paired Lymph Node Metastases of Triple Negative Breast Cancer

Authors:

Ming Li (mingli_1991@163.com)
Anqi Li (laqann@qq.com)
Shulin Zhou (sulina-214@163.com)
Yan Xu (xuyan11230321@163.com)
Yaoxing Xiao (yaoxing_xiao@163.com)
Rui Bi (br_fdcc@163.com)
Wentao Yang (yangwt2000@163.com)

Version: 2 Date: 28 Nov 2017

Author’s response to reviews:

Dear Editors,

We would like to thank you for your effort and for considering our manuscript “Heterogeneity of PD-L1 Expression in Primary Tumors and Paired Lymph Node Metastases of Triple Negative Breast Cancer” (BCAN-D-17-00104) for publication in “BMC Cancer”. Below you can find our response to the editorial comments.

Editor Comments:

1. Please add all authors’ email addresses before the abstract.

Response: We have added all authors’ email address before the abstract.

2. On uploading your revisions, please remove all tracking/highlighting and instead upload a single clean version of the manuscript.

Response: We have completed this part as the suggestions.
3. Please add a 'Conclusion' after the 'Discussion'.

Response: We have added this section “Conclusion” after the “Discussion”.

4. Please remove the titles/captions included in the figures and place them at the end of the main manuscript, after the References, in a section ‘Figures’ where you list the following information: * Figure # (e.g. Figure 1), * Title of Figure, and * Caption of Figure.

Response: we have removed the titles/captions included in the figures and placed them at the end of the main manuscript as the suggestions.

5. Please visit our submission guidelines and format your article accordingly: https://bmccancer.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

Response: We have added all the needed sections in declarations.

Yours Sincerely,

Wentao Yang on behalf of the authors.
Wentao Yang, M.D. &Ph.D.
Department of Pathology, Fudan University Shanghai Cancer Center
270 Dongan Road
Shanghai, 200032, P. R. China
Fax: 86-21- 64031696 Tel: 86-21-64031696
Email: yangwt2000@163.com